Author's response to reviews

Title: Barriers and facilitators to implement shared decision making in multidisciplinary sciatica care: a qualitative study

Authors:

Stefanie N Hofstede (s.n.hofstede@lumc.nl)
Perla J Marang-van de Mheen (p.j.marang-van_de_mheen@lumc.nl)
Manon M Wentink (m.wentink@sophiarevalidatie.nl)
Anne M Stiggelbout (a.m.stiggelbout@lumc.nl)
Carmen LA Vleggeert-Lankamp (c.l.a.m.vleggeert-lankamp@lumc.nl)
Thea PM Vliet Vlieland (t.p.m.vliet_vlieland@lumc.nl)
Leti van Bodegom-Vos (l.vanbodegom-vos@lumc.nl)

Version: 4 Date: 19 August 2013

Author's response to reviews: see over
Dear editor,

Thank you for the copy-edited version of our manuscript. We accepted all the suggested changes made by the copy-editor.

Concerning the question of the copy-editor; we changed: ‘Themes mentioned by both patients and professionals at the level of the social context are lack of/sufficient inter-professional collaboration and social influences of third parties.’ into ‘Themes mentioned by both patients and professionals at the level of the social context are (lack of) sufficient inter-professional collaboration and social influences of third parties.’ To clarify that this theme contains both barriers and facilitators.

We are looking forward to your reaction.

Yours sincerely,

On behalf of all authors,

Leti van Bodegom-Vos
corresponding author